Author’s response to reviews

Title: Randomised controlled trial of an intervention to improve parental knowledge and management practices of fever

Authors:

Maria Kelly (ehmkelly@yahoo.co.uk)
Laura Sahm (l.sahm@ucc.ie)
Suzanne McCarthy (s.mccarthy@ucc.ie)
Ronan O’Sullivan (ronan.osullivan@ucc.ie)
Aoife McGillicuddy (aoifecmc@eircom.net)
Frances Shiely (f.shiely@ucc.ie)

Version: 3 Date: 26 Oct 2019

Author’s response to reviews:

Dear Editor,
Thank you for accepting our manuscript subject to minor revisions. I have now addressed the 8 additional points.
1. Trial registration details have been amended as specified. Registration occurred retrospectively immediately after the recruitment and this has now been stated.
2. The abbreviations have been listed now in the section before the declarations
3. The statement in the consent for publication has been amended to say "Not applicable"
4. The additional statement has been added to the authors' contributions section.
5. The funding information has been removed from the Acknowledgements section.
6. I can confirm the images in Appendix 1 are our own and have not been taken from another source.
An "Additional Files" section has been added, as requested.
8. A clean manuscript version has now been uploaded.

Thank you again and I look forward to hearing from you about the final manuscript.

Sincerely yours,
Frances Shiely PhD